**Author’s response to reviews**

**Title:** Delayed HELLP syndrome in succession of preventive anticoagulant treatment-switch in a 41-year old patient with a history of recurrent assisted implantation failures: a case report.

**Authors:**

Katrin Mikolaiczik (katrinmik@t-online.de)

Marc Praetner (marc.pr@web.de)

Michael Rüth (michael.rueth@kliniken-nordoberpfalz.ag)

Karlheinz Mark (karlheinz.mark@kliniken-nordoberpfalz.ag)

**Version:** 1 **Date:** 07 Nov 2018

**Author’s response to reviews:**

#1: Terms like ours, we etc. were removed. I could shorten the manuscript. The number of references though should underline the argumentation and should be a source of information for interested gynaecologists. Therefore no shortening was done on the number of references.